Tutorial #4 (for week 7) (Feb 18, 2003)

15th February 2003

The following questions will be considered during the tutorial sessions on Tuesday...

Q1: Write a minimal Tk application that shows a textfile in a scrollable listbox. Q2: Write a minimal Tk application using the canvas, that shows a bouncing ball, with a scale that varies the speed of the ball.

Q3: Research the Tk geometry managers. Briefly describe each one, and outline its use. Q4: Write Tcl code that creates a new Tcl command structure - the repeat x until y command.